Author's response to reviews

Title: Case Report: Acute Bowel Obstruction with an Isolated Transition Point in Peritoneal Dialysis Patients; A Presentation of Encapsulating Peritoneal Sclerosis?

Authors:

Sassan Ghazan-Shahi (s_ghazanshahi@yahoo.ca)
Joanne M Bargman (joanne.bargman@uhn.ca)

Version: 4 Date: 27 November 2015

Author's response to reviews: see over
November 26, 2015

Title:
Case Report: Acute Bowel Obstruction with an Isolated Transition Point in Peritoneal Dialysis Patients; A Presentation of Encapsulating Peritoneal Sclerosis?

Authors:
Sassan Ghazan-Shahi, MD FRCPC, Joanne M Bargman, MD FRCPC

Dear Editor,
Thank you very much for considering our manuscript for publication in the BMC Nephrology Journal.

As requested we have done the following revisions:

1. Edited the CARE checklist to include the page numbers where each item appears rather than a tick
2. Formatted the AC to mentions each author specifically by their initials.
3. Included a Conclusions section
4. Added “Case Report” to the title, as per CARE checklist requirement.

Please do not hesitate to contact us should you require any further information. We will forward to hearing from you. Thank you.

Sincerely,
Sassan Ghazan-Shahi, MD FRCPC
Joanne M Bargman, MD FRCPC